Author's response to reviews

Title: Incidence of and risk factors for Motor Neurone Disease in UK women: A prospective study

Authors:

Pat Doyle (pat.doyle@lshtm.ac.uk)
Anna Brown (anna.brown@ceu.ox.ac.uk)
Valerie Beral (pa.valerie.beral@ceu.ox.ac.uk)
Gillian Reeves (gillian.reeves@ceu.ox.ac.uk)
Jane Green (jane.green@ceu.ox.ac.uk)

Version: 4 Date: 23 March 2012

Author’s response to reviews:

Dear Editors,

Thank you for sending the reviewers comments on this paper. Reviewer 1 thought the paper acceptable and had no suggestions for revision.

Reviewer 2 commented on text in the Discussion relating to published evidence on the link between smoking and MND/ALS. We have responded to this reviewer’s comment by including new text and two further references relating to this point (in red type in the revised manuscript). We hope that discussion on this point is now improved.

We look forward to hearing from you.

Yours sincerely,

Pat Doyle